Author’s response to reviews

Title: Sequential E2 Levels not Ovarian Maximal Diameter Estimates were Correlated with Outcome of Cetrotide Therapy for Management of Women at High-risk of Ovarian Hyperstimulation Syndrome: A randomised Controlled Study

Authors:
Khalid Salama (dr.khalid_sleem@yahoo.com)
Hesham Abo Ragab (ahmed-1919@windowslive.com)
Mohammed El Sherbiny (Mohfarag171@gmail.com)
Ali Elnaby (alimalim528@rocketmail.com)
Ibrahim Souidan (ebrahim.swedan@yahoo.com)

Version: 3 Date: 31 Oct 2017

Author’s response to reviews:

1. After your abstract please include ALL trial registration details. This includes: name of registry, registration number and date of registration Comment: added (abstract-page 2-line 43, 44)

2. Please also include this information in your "ethical approval and consent to participate" section of your declarations Comment: added (ethical approval and consent to participate page 18-line 362-363)

3. Please remove the initials "MD" from your author contributions section as this does not correspond to any of the authors on this paper Comment: removed (author contributions page 17-line 346-353)

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Comment: uploaded